Author's response to reviews

Title: Behaviour change in perinatal care practices among rural women exposed to a women's group intervention in Nepal [ISRCTN31137309]

Authors:

Angela M Wade (awade@ich.ucl.ac.uk)
David Osrin (d.osrin@ich.ucl.ac.uk)
Bhim P Shrestha (mirajnk@ntc.net.np)
Aman Sen (miramak@ntc.net.np)
Joanna Morrison (j.morrison@ich.ucl.ac.uk)
Kirti M Tumbahangphe (miramak@ntc.net.np)
Dharma S Manandhar (dsm@healthnet.org.np)
Anthony ML Costello (a.costello@ich.ucl.ac.uk)

Version: 2 Date: 8 February 2006

Author's response to reviews:

The trial registration number has been included immediately following the title of the paper. Previously this was given at the foot of the title page. We apologise for omitting to place correctly initially.

Please let us know if more information is required/this is not satisfactory.
Thank you.
Yours sincerely
A. M. Wade